Author's response to reviews

Title: Cyclooxygenase-2 overexpression is common in serrated and non-serrated colorectal adenoma, but uncommon in hyperplastic polyp and sessile serrated polyp/adenoma

Authors:

Takako Kawasaki (takako_kawasaki@dfci.harvard.edu)
Katsuhiko Nosho (katsuhiko_nosho@dfci.harvard.edu)
Mutsuko Ohnishi (mutsuko_ohnishi@dfci.harvard.edu)
Yuko Suemoto (yuyupon26@yahoo.co.jp)
Jonathan N Glickman (jglickman@partners.org)
Andrew T Chan (achan@partners.org)
Gregory J Kirkner (gregory.kirkner@channing.harvard.edu)
Mari Mino-Kenudson (mmino@partners.org)
Charles S Fuchs (cfuchs@dfci.harvard.edu)
Shuji Ogino (shuji_ogino@dfci.harvard.edu)

Version: 5 Date: 28 January 2008

Author's response to reviews:

We have changed the manuscript according to formatting instructions.